

Received May 9, 2019, accepted June 17, 2019, date of publication June 20, 2019, date of current version July 9, 2019. *Digital Object Identifier* 10.1109/ACCESS.2019.2924034

Intelligent Reflecting Surface: A Programmable Wireless Environment for Physical Layer Security

JIE CHEN^{(D1,2}, (Student Member, IEEE), YING-CHANG LIANG^(D2), (Fellow, IEEE), YIYANG PEI^(D3), (Member, IEEE), AND HUAYAN GUO^(D1,2), (Member, IEEE)

¹National Key Laboratory of Science and Technology on Communications, University of Electronic Science and Technology of China (UESTC), Chengdu 611731, China

²Center for Intelligent Networking and Communications (CINC), University of Electronic Science and Technology of China (UESTC), Chengdu 611731, China ³Infocomm Technology Cluster, Singapore Institute of Technology, Singapore 138683

Corresponding author: Ying-Chang Liang (liangyc@ieee.org)

This work was supported in part by the National Natural Science Foundation of China under Grant 61631005, Grant U1801261, and Grant 61571100.

ABSTRACT In this paper, we introduce an *intelligent reflecting surface* (IRS) to provide a programmable wireless environment for physical layer security. By adjusting the reflecting coefficients, the IRS can change the attenuation and scattering of the incident electromagnetic wave so that it can propagate in the desired way toward the intended receiver. Specifically, we consider a downlink *multiple-input single-output* (MISO) broadcast system, where the *base station* (BS) transmits independent data streams to multiple legitimate receivers and keeps them secret from multiple eavesdroppers. By jointly optimizing the beamformers at the BS and reflecting coefficients at the IRS, we formulate a minimum-secrecy-rate maximization problem under various practical constraints on the reflecting coefficients. The constraints capture the scenarios of both continuous and discrete reflecting coefficients of the reflecting elements. Due to the non-convexity of the formulated problem, we propose an efficient algorithm based on the alternating optimization and the path-following algorithm to solve it in an iterative manner. Besides, we show that the proposed algorithm can converge to a local (global) optimum. Furthermore, we develop two suboptimal algorithms with some forms of closed-form solutions to reduce computational complexity. Finally, the simulation results validate the advantages of the introduced IRS and the effectiveness of the proposed algorithms.

INDEX TERMS Intelligent reflecting surface, programmable wireless environment, physical layer security, beamforming.

I. INTRODUCTION

A variety of wireless technologies have been proposed to enhance the spectrum- and energy-efficiency due to the tremendous growth in the number of communication devices, such as *multiple-input multiple-output* (MIMO) [1], cooperative communications [2], *cognitive radio* (CR) [3] and so on. However, these techniques only focus on the signal processing at the transceiver to adapt the changes of the wireless environment, but cannot eliminate the negative effects caused by the uncontrollable electromagnetic wave propagation environment [4], [5]. Recently, *intelligent reflecting surface* (IRS) has been proposed as a promising technique due to its capability to achieve high spectrum-/energy-efficiency through controlling the wireless propagation environment [6]. Specifically, IRS is a uniform planar array consisting of a large number of composite material elements, each of which can adjust the reflecting coefficients (i.e., phase or amplitude) of the incident electromagnetic wave and reflect it passively. Hence, by smartly adjusting the reflecting coefficients with a preprogrammed controller, the IRS can change the attenuation and scattering of the incident electromagnetic wave so that it can propagate in the desired way before reaching the intended receiver, which is called as programmable and controllable wireless environment. This also inspires us to design the communication systems by jointly considering the signal

The associate editor coordinating the review of this manuscript and approving it for publication was Matti Hämäläinen.

processing at the transceiver and the optimization of the electromagnetic wave propagation in the wireless environment.

Compared with the existing related techniques, i.e., traditional reflecting surfaces [7], amplify-and-forward (AF) relay [8], active intelligent surface [9], and backscatter communication [10]-[12], IRS has the following advantages. Firstly, IRS can reconfigure the reflecting coefficients in real time thanks to the recent breakthrough on *micro-electrical*mechanical systems (MEMS) and composite material [5], [6] while the traditional reflecting surface only has fixed reflecting coefficients. Secondly, IRS is a green and energy-efficient technique which reflects the incident signal passively without additional energy consumption while the AF relay and the active intelligent surface require active radio frequency (RF) components. Thirdly, although both the IRS and the backscatter communication make use of passive communications, IRS can be equipped with a large number of reflecting elements while backscatter devices are usually equipped with a single/few antenna(s) due to the limitations of complexity and cost [13]. Besides, IRS only attempts to assist the transmission of the signals between the intended transmitter and receiver pair with no intention for its own information transmission while backscatter communication needs to support the information transmission of the backscatter device [14], [15].

Due to the significant advantages, IRS has been introduced into various wireless communication systems. Specifically, [16]-[20] consider a downlink single user multipleinput single-output (MISO) system assisted by the IRS. In [16], both centralized and distributed algorithms were developed to maximize the signal-to-noise ratio (SNR) of the desired signals considering perfect channel state information (CSI). Then, in [17], the effect of the reflecting coefficients on the ergodic capacity was investigated by considering statistical CSI. Moreover, since achieving continuous reflecting coefficients on the reflecting elements is costly in practice due to the hardware limitation, the SNR maximization problem and transmitter power minimization problem were studied in [18]–[20] by considering discrete reflecting coefficients on the reflecting elements. As for a downlink multi-user MISO system [21]-[23], the spectrum-/ energy-efficiency problem under the individual signalto-interference-plus-noise ratio (SINR) constraints was investigated in [21] and [22] considering continuous or discrete reflecting coefficients on the reflecting elements. In addition, the minimum-SINR maximization problem was formulated in [23] by considering the two cases where the channel matrix between the transmitter and the IRS is of rankone and of full-rank.

Furthermore, physical layer security is a fundamental issue in wireless communications [24]. The basic wiretap channel introduced by Wyner [25] consists of one transmitter, one legitimate receiver, and one eavesdropper. Then, the basic wiretap channel has been extended to broadcast channels [26], Gaussian channels [27], compound wiretap channels [28], and so on. It is worth noting that, in order to ensure secret communications, the transmission rate in the wiretap channel should be lower than the secrecy capacity of the channel. Thus, MIMO beamforming techniques were further introduced to improve the secrecy capacity (improving SNR of legitimate receivers and suppressing SNR of eavesdroppers) [29]–[32]. Specifically, both power minimization and secrecy rate maximization were studied in [30] in a single user/eavesdropper MIMO systems considering both perfect and imperfect CSI. Then, the minimum-secrecy-rate of a single-cell multi-user MISO system was studied in [31] with a minimum harvested energy constraint, and it was further extended to a multi-cell network in [32].

However, consider the special case when the legitimate receivers and the eavesdroppers are in the same directions to the transmitter. In this case, the channel responses of the legitimate receivers will be highly correlated with those of the eavesdroppers. The beamformers proposed in [29]–[32] to maximize the SNR of legitimate receivers will also maximize the SNR of eavesdroppers. Hence, it is intractable to guarantee the secret communications with the use of beamforming only at the transceivers. Hence, we want to explore the use of the IRS to provide additional communication links so as to increase the SNR at the legitimate receivers while suppressing the SNR at the eavesdroppers. Hopefully, this will create an effect as if the confidential data streams can bypass the eavesdroppers and reach the legitimate receivers, as shown in Fig. 1, and thus the secrecy rate will be improved.

Motivated by the above reasons, in this paper, we study a programmable wireless environment for physical layer security to achieve high-efficiency secret communication. Specifically, we consider a downlink MISO broadcast system where the *base station* (BS) transmits multiple independent confidential data streams to each legitimate receivers and keeps them secret from the eavesdroppers through the assistance of the IRS. The contributions of the paper are summarized as follows:

- To the best of our knowledge, this is the first work to explore the use of the IRS to enhance the physical layer secret communication. Particularly, we jointly optimize the beamformers at the BS and the reflecting coefficients at the IRS to maximize the minimum-secrecy-rate under various practical constraints on the reflection coefficients. The constraints capture both the continuous and discrete reflecting coefficients of the reflecting elements on the IRS. However, the objective function is not jointly concave with respect to both the beamformers and the reflecting coefficients, and even worse, they are coupled together. Hence, the formulated problem is non-convex, which is hard to solve and may require high complexity to obtain the optimal solutions.
- We solve the formulated problem efficiently in an iterative manner by developing alternating optimization based path-following algorithm [33], [34]. Specifically, we use the path-following algorithm to handle the non-concavity of the objective function and apply the

alternating optimization to deal with the coupled optimization variables. Besides, we prove that the proposed algorithm is guaranteed to converge to a local (global) optimum and the corresponding solution will converge to a *Karush-Kuhn-Tucker* (KKT) point finally.

- To further reduce the computational complexity, we develop two suboptimal algorithms to solve the formulated problem for two cases. For the first case with one legitimate receiver and one eavesdropper, we develop an alternating optimization method to solve the formulated problem in an iterative manner, but in each iteration we provide the closed-form solutions, which leads the algorithm to be low complexity. For the second case with multiple legitimate receivers and eavesdroppers, we develop a heuristic closed-form solution based on *zero-forcing* (ZF) beamforming, which further reduces the computational complexity.
- Finally, the simulation results validate the advantages of the introduced IRS and also show the effectiveness of the proposed algorithms.

The rest of this paper is organized as follows: Section II introduces the system model of the downlink MISO broadcast system with multiple eavesdroppers. Section III formulates the minimum-secrecy-rate maximization problem. Section IV develops an efficient algorithm to solve the formulated problem and Section V provides two low-complexity suboptimal algorithms to solve it in two cases, respectively. Section VI shows the simulation results to evaluate the performances of the proposed algorithms. Finally, Section VII concludes the paper.

The notations used in this paper are listed as follows. The scalar, vector, and matrix are lowercase, bold lowercase, and bold uppercase, i.e., a, \mathbf{a} , and \mathbf{A} , respectively. $(\cdot)^T$, $(\cdot)^H$, Tr (\cdot) and $\Re(\cdot)$ denote transpose, conjugate transpose, trace, and real dimension, respectively. $C\mathcal{N}(\mu, \sigma^2)$ denotes the distribution of a *circularly symmetric complex Gaussian* (CSCG) random variable with mean μ and variance σ^2 . $\mathbb{C}^{x \times y}$ and $\mathbb{R}^{x \times y}$ denote the space of $x \times y$ complex/real matrices. $I_K \in \mathbb{R}^{K \times K}$ is the identify matrix, $\mathbf{1}_K = [1, \cdots, 1]^T \in \mathbb{R}^{K \times 1}$, and $(a)^+ = \max(0, a)$.

II. SYSTEM MODEL

As shown in Fig. 1, we consider a programmable downlink MISO broadcast system which consists of one BS, one IRS, K legitimate receivers, denoted as B_1, \dots, B_K , and N active eavesdroppers, denoted as E_1, \dots, E_N . The BS and the IRS are equipped with M antennas and L reflecting elements, respectively, while the legitimate receivers and eavesdroppers are all equipped with a single antenna each. The BS sends K independent confidential data streams with one stream for each of the K legitimate receivers over the same frequency band, simultaneously. At the same time, the unauthorized eavesdroppers are trying to eavesdrop any of the data streams, independently.

Consider the special case when the legitimate receivers and the eavesdroppers are in the same directions to the BS.



FIGURE 1. A programmable downlink MISO broadcast system with one IRS and multiple eavesdroppers.

In this case, the channel responses of the legitimate receivers will be highly correlated with those of the eavesdroppers. As aforementioned, it is intractable to guarantee the secret communications with the use of beamforming only at the transceivers. Hence, we want to explore the use of the IRS to provide additional communication links so as to increase the SNR at the legitimate receivers while suppressing the SNR at the eavesdroppers. Hopefully, this will create an effect as if the confidential data streams can bypass the eavesdroppers and reach the legitimate receivers, and thus the secrecy rate will be improved. In this paper, we are interested in obtaining the performance limit of such a system. Hence, similarly to [16] and [21], we assume that the CSI of all the channels are perfectly known at the BS.¹ In practical systems where such CSI cannot be obtained perfectly, the results derived in this paper can be considered as the performance upper bound. Note that the optimization (in terms of beamformers and reflecting coefficients) of the system to be presented in the subsequent sections is done at the BS and that the optimized reflecting coefficients are transmitted to the IRS to reconfigure the corresponding reflecting elements accordingly.

A. CHANNEL MODEL

The baseband equivalent channel responses from the BS to the IRS, from the BS to B_k , from the BS to E_n , from the IRS to B_k , and from the IRS to E_n are denoted by $F \in \mathbb{C}^{L \times M}$, $\mathbf{h}_{d,k}^H \in \mathbb{C}^{1 \times M}$, $\mathbf{g}_{d,n}^H \in \mathbb{C}^{1 \times M}$, $\mathbf{h}_{r,k}^H \in \mathbb{C}^{1 \times L}$, and $\mathbf{g}_{r,n}^H \in \mathbb{C}^{1 \times L}$, respectively, with $1 \le k \le K$ and $1 \le n \le N$. Specifically, without loss of generality, we adopt a Rician fading channel model, which consists of LoS and

¹In practice, the optimization processing only requires the CSIs of the composite channels, i.e., H_k and G_n , defined in (15) and (16). Hence, we can first turn off the IRS and enable the users to send orthogonal pilot sequences to the BS for estimating the last rows of H_k and G_n . Next, we can turn on each reflecting element on the IRS successively in *L* time slots and keep the other reflecting elements closed. Then, the users send orthogonal pilot sequences to the BS in the each time slot. Finally, the first *L* rows in H_k and G_n can be estimated by subtracting the last rows of H_k and G_n from the estimated channel gains.

non-LoS (NLoS) components, i.e.,

$$\boldsymbol{h} = \sqrt{\frac{\kappa_{\boldsymbol{h}}}{\kappa_{\boldsymbol{h}} + 1}} \boldsymbol{h}^{\text{LoS}} + \sqrt{\frac{1}{\kappa_{\boldsymbol{h}} + 1}} \boldsymbol{h}^{\text{NLoS}}, \qquad (1)$$

with $h \in \mathcal{H} = \{F, h_{d,k}, h_{r,k}, g_{d,n}, g_{r,n}\}$, where κ_h, h^{LoS} , and h^{NLoS} are the Rician factor, LoS components, and NLoS components of channel h, respectively. The NLoS components h^{NLoS} are i.i.d. complex Gaussian distributed with zero mean and unit variance. We define a vector $a_X(\vartheta) = [1, e^{j\frac{2\pi d}{\lambda}\sin\vartheta}, \cdots, e^{j\frac{2\pi d}{\lambda}(X-1)\sin\vartheta}]^T$, where d is the antenna element separation, λ is the carrier wavelength, X is the dimension of the vector and ϑ is the angle, which can be interpreted as either *angle of departure* (AoD) or *angle of arrival* (AoA) depending on the context. We set $d/\lambda = 1/2$ for simplicity. Hence, the LoS components in (1) can be modeled as

$$\boldsymbol{h}_{d,k}^{\text{LoS}} = \boldsymbol{a}_{M}(\vartheta_{d,k}) \text{ and } \boldsymbol{h}_{r,k}^{\text{LoS}} = \boldsymbol{a}_{L}(\vartheta_{r,k}), \text{ for } 1 \le k \le K,$$
(2)

$$\boldsymbol{g}_{d,n}^{\text{LoS}} = \boldsymbol{a}_{M}(\tilde{\vartheta}_{d,n}) \text{ and } \boldsymbol{g}_{r,n}^{\text{LoS}} = \boldsymbol{a}_{L}(\tilde{\vartheta}_{r,n}), \quad \text{for } 1 \le n \le N,$$
(3)

$$F^{\text{LoS}} = a_L \left(\vartheta^{\text{AoA}} \right) a_M^H \left(\vartheta^{\text{AoD}} \right), \tag{4}$$

where $\vartheta_{d,k}$, $\vartheta_{r,k}$, $\tilde{\vartheta}_{d,n}$, $\tilde{\vartheta}_{r,n}$ are the AoA or AoD of a signal from the BS to B_k , from the IRS to B_k , from the BS to E_n , and from the IRS to E_n , respectively. ϑ^{AoD} and ϑ^{AoA} are the AoD from the BS and the AoA to the IRS, respectively.

B. REFLECTING COEFFICIENT MODEL

The reflecting coefficient channel of the IRS [16] is given by $\Theta = \operatorname{diag}(\theta) \in \mathbb{C}^{L \times L}$ with $\theta = [\theta_1, \theta_2, \dots, \theta_L]^T \in \mathbb{C}^{L \times 1}$ and $\theta_l \in \Phi$ for $1 \leq l \leq L$, where diag(·) denotes a diagonal matrix whose diagonal elements are given by the corresponding vector and Φ denotes the set of reflecting coefficients of the IRS. In this paper, we consider the following three different sets of reflecting coefficients, which lead to three different constraints for the reflecting coefficients.

• Continuous Reflecting Coefficients: In this scenario, we further consider two detailed setups with the optimized or constant amplitude. Specifically, the reflecting coefficient set for the optimized amplitude with continuous phase-shift is denoted by

$$\mathbf{\Phi}_1 = \left\{ \theta_l \left| |\theta_l|^2 \le 1 \right\},\tag{5}$$

and the reflecting coefficient set for the constant amplitude with continuous phase-shift is denoted by

$$\mathbf{\Phi}_2 = \left\{ \theta_l \middle| \theta_l = e^{j\varphi_l}, \varphi_l \in [0, 2\pi) \right\}.$$
(6)

• *Discrete Reflecting Coefficients:* In this scenario, the reflecting coefficient set has constant amplitude and discrete phase-shift, which is given by

$$\Phi_3 = \left\{ \theta_l \left| \theta_l = e^{j\varphi_l}, \varphi_l \in \left\{ 0, \frac{2\pi}{Q}, \cdots, \frac{2\pi(Q-1)}{Q} \right\} \right\}, \quad (7)$$

where Q is the number of reflecting coefficient values of the reflecting elements on the IRS.

Note that, it is costly in practice to achieve continuous reflecting coefficient on the reflecting elements due to the hardware limitation. Hence, applying the discrete reflecting coefficient on the reflecting elements, i.e., Φ_3 , is more practical than applying the continuous reflecting coefficients, i.e., Φ_1 and Φ_2 . But, it is also important to investigate the system performance with Φ_1 and Φ_2 since it serves as the upper bound to that with Φ_3 .

C. SIGNAL MODEL

Let s_k be the confidential message dedicated to B_k . It is assumed that all messages transmitted are CSCG, i.e., $s_k \sim C\mathcal{N}(0, 1)$ for $1 \le k \le K$. Then, the signal transmitted from the BS can be expressed as

$$\mathbf{x} = \sum_{k=1}^{K} \mathbf{w}_k s_k,\tag{8}$$

where w_k is the downlink beamforming vector for s_k . The received signals at B_k and eavesdropped by E_n can be expressed as

$$y_k^{\mathbf{B}} = \left[\boldsymbol{h}_{r,k}^{H} \boldsymbol{\Theta} \boldsymbol{F} + \boldsymbol{h}_{d,k}^{H} \right] \sum_{i=1}^{K} \boldsymbol{w}_i x_i + u_k^{\mathbf{B}}, 1 \le k \le K, \quad (9)$$

$$y_n^{\rm E} = \left[\boldsymbol{g}_{r,n}^{H} \boldsymbol{\Theta} \boldsymbol{F} + \boldsymbol{g}_{d,n}^{H} \right] \sum_{i=1}^{K} \boldsymbol{w}_i x_i + \boldsymbol{u}_n^{\rm E}, 1 \le n \le N, \quad (10)$$

respectively, where $u_k^{\rm B}$ and $u_n^{\rm E}$ are the received noises at B_k and E_n , respectively. It is assumed that all noises are Gaussian distributed with zero mean, i.e., $u_k^{\rm B} \sim C\mathcal{N}(0, \sigma_k^2)$ and $u_k^{\rm E} \sim C\mathcal{N}(0, \delta_n^2)$, respectively.

According to (9), the achievable transmission rate of the k-th confidential message received at B_k can be written as

$$R_{k}^{\mathrm{B}} = \ln\left(1 + \frac{\left|(\boldsymbol{h}_{r,k}^{H}\boldsymbol{\Theta}\boldsymbol{F} + \boldsymbol{h}_{d,k}^{H})\boldsymbol{w}_{k}\right|^{2}}{\sum_{i \neq k}^{K}\left|(\boldsymbol{h}_{r,k}^{H}\boldsymbol{\Theta}\boldsymbol{F} + \boldsymbol{h}_{d,k}^{H})\boldsymbol{w}_{i}\right|^{2} + \sigma_{k}^{2}}\right).$$
 (11)

According to (10), if E_n attempts to eavesdrop the k-th confidential message, the achievable wiretapped rate of the k-th message received at E_n can be written as

$$R_{k,n}^{\mathrm{E}} = \ln \left(1 + \frac{\left| (\boldsymbol{g}_{r,n}^{H} \boldsymbol{\Theta} \boldsymbol{F} + \boldsymbol{g}_{d,n}^{H}) \boldsymbol{w}_{k} \right|^{2}}{\sum_{i \neq k}^{K} \left| (\boldsymbol{g}_{r,n}^{H} \boldsymbol{\Theta} \boldsymbol{F} + \boldsymbol{g}_{d,n}^{H}) \boldsymbol{w}_{i} \right|^{2} + \delta_{n}^{2}} \right).$$
(12)

Since each eavesdropper can eavesdrop any of the *K* confidential messages, the achievable secrecy rate (in nats/sec/Hz) for transmitting s_k to B_k and keeping it confidential from all the *N* eavesdroppers should be the minimum-secrecy-rate among B_k and E_n for $1 \le n \le N$, which is given by [32]

$$C_k = \min_{\forall n} \left\{ R_k^{\rm B} - R_{k,n}^{\rm E} \right\}.$$
(13)

III. PROBLEM STATEMENT

A. PROBLEM FORMULATION

In this paper, we attempt to jointly optimize the beamforming vector, i.e., $W = [w_1, \dots, w_k] \in \mathbb{C}^{M \times K}$, and reflecting coefficients, i.e., θ , to maximize the minimum-secrecy-rate among all the legitimate receivers. Mathematically, the optimized problem can be generally formulated as

$$(\mathbf{P}1) : \max_{\boldsymbol{W},\boldsymbol{\theta}} \min_{\forall k} C_k$$

s.t. $\sum_{k=1}^{K} \|\boldsymbol{w}_k\|^2 \le P,$ (14a)
 $\theta_l \in \boldsymbol{\Phi}, 1 \le l \le L,$ (14b)

where *P* denotes the maximum transmit power at the BS and Φ may be set as Φ_1 , Φ_2 , and Φ_3 , respectively.

B. PROBLEM TRANSFORMATION

 $(\mathbf{P}1)$ is hard to solve due to the non-concave objective function. In order to find the solution of $(\mathbf{P}1)$ efficiently, we will transform it into the following equivalent formulation.

To begin with, denoting
$$\boldsymbol{H}_{k} = \begin{bmatrix} \operatorname{diag}(\boldsymbol{h}_{r,k}^{H})\boldsymbol{F} \\ \boldsymbol{h}_{d,k}^{H} \end{bmatrix} \in \mathbb{C}^{(L+1)\times M}$$
, and $\boldsymbol{G}_{n} = \begin{bmatrix} \operatorname{diag}(\boldsymbol{g}_{r,n}^{H})\boldsymbol{F} \\ \boldsymbol{g}_{d,n}^{H} \end{bmatrix} \in \mathbb{C}^{(L+1)\times M}$, we have

$$\left| (\boldsymbol{h}_{r,k}^{H} \boldsymbol{\Theta} \boldsymbol{F} + \boldsymbol{h}_{d,k}^{H}) \boldsymbol{w}_{k} \right|^{2} = \left| \boldsymbol{v}^{H} \boldsymbol{H}_{k} \boldsymbol{w}_{k} \right|^{2}, \qquad (15)$$

$$\left| (\boldsymbol{g}_{r,n}^{H} \boldsymbol{\Theta} \boldsymbol{F} + \boldsymbol{g}_{d,n}^{H}) \boldsymbol{w}_{k} \right|^{2} = \left| \boldsymbol{v}^{H} \boldsymbol{G}_{n} \boldsymbol{w}_{k} \right|^{2}, \qquad (16)$$

where $\boldsymbol{v} = [v_1, v_2, \cdots, v_{L+1}]^T = [\boldsymbol{\theta}; 1] \in \mathbb{C}^{1 \times (L+1)}$. Then, R_k^{B} in (11) and $R_{k,n}^{\text{E}}$ in (12) can be rewritten as

$$\boldsymbol{R}_{k}^{\mathrm{B}} = \ln\left(1 + \frac{\left|\boldsymbol{\nu}^{H}\boldsymbol{H}_{k}\boldsymbol{w}_{k}\right|^{2}}{b_{k}\left(\boldsymbol{W},\boldsymbol{\nu}\right)}\right) \stackrel{\Delta}{=} f_{k}^{\mathrm{B}}\left(\boldsymbol{W},\boldsymbol{\nu}\right), \quad (17)$$

$$R_{k,n}^{\mathrm{E}} = \ln\left(1 + \frac{|\mathbf{v}^{H} \mathbf{G}_{n} \mathbf{w}_{k}|^{2}}{q_{k,n}(\mathbf{W}, \mathbf{v})}\right) \stackrel{\Delta}{=} f_{k,n}^{\mathrm{E}}(\mathbf{W}, \mathbf{v}), \quad (18)$$

where $b_k(\mathbf{W}, \mathbf{v}) = \sum_{i \neq k}^{K} |\mathbf{v}^H \mathbf{H}_k \mathbf{w}_i|^2 + \sigma_k^2$ and $q_{k,n}(\mathbf{W}, \mathbf{v}) = \sum_{i \neq k}^{K} |\mathbf{v}^H \mathbf{G}_n \mathbf{w}_i|^2 + \delta_n^2$. Thus, it is straightforward to know that (**P**1) can be transformed into the following equivalent form:

$$(\mathbf{P2}): \max_{\boldsymbol{W},\boldsymbol{\nu}} \mathcal{R}(\boldsymbol{W},\boldsymbol{\nu}) \stackrel{\Delta}{=} \min_{\forall k,\forall n} \left\{ f_k^{\mathrm{B}}(\boldsymbol{W},\boldsymbol{\nu}) - f_{k,n}^{\mathrm{E}}(\boldsymbol{W},\boldsymbol{\nu}) \right\}$$

s.t. $v_l \in \boldsymbol{\Phi}, 1 \le l \le L, v_{L+1} = 1,$ (19a)
(14a).

However, the transformed problem (**P**2) is still hard to solve since $\mathcal{R}(W, v)$ is not jointly concave with respect to W and v, and even worse, they are coupled together. In the next section, we will develop an iterative algorithm to solve (**P**2) efficiently.

IV. MINIMUM-SECRECY-RATE MAXIMIZATION

In this section, we will propose two techniques to jointly solve the above challenging problem. Firstly, we apply the path-following algorithm to handle the non-concavity of the objective function. Then, we apply the alternating optimization technique to deal with the coupled optimization variables. Finally, we analyze the convergence of the proposed algorithm.

A. PATH-FOLLOWING ALGORITHM DEVELOPMENT

In this part, we will develop path-following iterative algorithm to solve (**P**2) with the non-concave objective function, i.e., $\mathcal{R}(W, v)$. In particular, the basic idea of the path-following is to follow a solution path of a family of the approximated problems of (**P**2). For example, $\mathcal{R}(W, v)$ is approximated by a concave lower bound function, which is obtained by applying linearly interpolating between the non-concave term $f_k^{\text{B}}(W, v)$ and the non-convex term $f_{k,n}^{\text{E}}(W, v)$, respectively. Specifically, the approximated problem has a local (global) optimal value and can be increased in each iteration, which finally leads to a local (global) optimal solution of (**P**2) [33].

To begin with, let $(\mathbf{W}^{(t)}, \mathbf{v}^{(t)})$ denote the solution of (**P**2) in the *t*-th iteration. Then, in order to find the concave lower bound function of $\mathcal{R}(\mathbf{W}, \mathbf{v})$ to develop path-following algorithm, we can fist find the lower bound function of $f_k^{\rm B}(\mathbf{W}, \mathbf{v})$ and the upper bound function of $f_{k,n}^{\rm E}(\mathbf{W}, \mathbf{v})$ at $(\mathbf{W}^{(t)}, \mathbf{v}^{(t)})$. The details are given in the following lemma.

Lemma 1: The lower bound function of $f_k^{\rm B}(W, v)$ and the upper bound function of $f_{k,n}^{\rm E}(W, v)$ at $(W^{(t)}, v^{(t)})$ in the (t + 1)-th iteration of path-following algorithm are given by

$$f_{k}^{\mathbf{B}}(\mathbf{W}, \mathbf{v}) \geq f_{k}^{\mathbf{B}}(\mathbf{W}^{(t)}, \mathbf{v}^{(t)}) + 2 \frac{\Re\left\{\left(\mathbf{w}_{k}^{(t)}\right)^{H} \mathbf{H}_{k}^{H} \mathbf{v}^{(t)}(\mathbf{v}^{H} \mathbf{H}_{k} \mathbf{w}_{k})\right\}}{b_{k}(\mathbf{W}^{(t)}, \mathbf{v}^{(t)})} \\ - \frac{\left|\left(\mathbf{v}^{(t)}\right)^{H} \mathbf{H}_{k} \mathbf{w}_{k}^{(t)}\right|^{2}}{b_{k}(\mathbf{W}^{(t)}, \mathbf{v}^{(t)})(b_{k}(\mathbf{W}^{(t)}, \mathbf{v}^{(t)}) + \left|\left(\mathbf{v}^{(t)}\right)^{H} \mathbf{H}_{k} \mathbf{w}_{k}^{(t)}\right|^{2})} \\ \times \left(\left|\mathbf{v}^{H} \mathbf{H}_{k} \mathbf{w}_{k}\right|^{2} + b_{k}(\mathbf{W}, \mathbf{v})\right) - \frac{\left|\left(\mathbf{v}^{(t)}\right)^{H} \mathbf{H}_{k} \mathbf{w}_{k}^{(t)}\right|^{2}}{b_{k}(\mathbf{W}^{(t)}, \mathbf{v}^{(t)})} \\ \stackrel{\Delta}{=} f_{k}^{\mathbf{B}}(\mathbf{W}, \mathbf{v}; \mathbf{W}^{(t)}, \mathbf{v}^{(t)}), \tag{20}$$

$$\begin{aligned}
f_{k,n}^{\mathrm{E}}(\boldsymbol{W},\boldsymbol{\theta}) &\leq f_{k,n}^{\mathrm{E}}(\boldsymbol{W}^{(t)},\boldsymbol{v}^{(t)}) + (1 + \frac{|\boldsymbol{v}^{(t')} \cdot \boldsymbol{v}^{(n_{k}}||}{q_{k,n}(\boldsymbol{W}^{(t)},\boldsymbol{v}^{(t)})})^{-1} \\
&\times (\frac{|\boldsymbol{v}^{H}\boldsymbol{G}_{n}\boldsymbol{w}_{k}|^{2}}{q_{k,n}(\boldsymbol{W},\boldsymbol{v})} - \frac{|(\boldsymbol{v}^{(t)})^{H}\boldsymbol{G}_{n}\boldsymbol{w}_{k}^{(t)}|^{2}}{q_{k,n}(\boldsymbol{W}^{(t)},\boldsymbol{v}^{(t)})}) \\
&\leq f_{k,n}^{\mathrm{E}}(\boldsymbol{W}^{(t)},\boldsymbol{v}^{(t)}) + (1 + \frac{|(\boldsymbol{v}^{(t)})^{H}\boldsymbol{G}_{n}\boldsymbol{w}_{k}^{(t)}|^{2}}{q_{k,n}(\boldsymbol{W}^{(t)},\boldsymbol{v}^{(t)})})^{-1} \\
&\times (\frac{|\boldsymbol{v}^{H}\boldsymbol{G}_{n}\boldsymbol{w}_{k}|^{2}}{q_{k,n}(\boldsymbol{W},\boldsymbol{v};\boldsymbol{W}^{(t)},\boldsymbol{v}^{(t)})} - \frac{|(\boldsymbol{v}^{(t)})^{H}\boldsymbol{G}_{n}\boldsymbol{w}_{k}^{(t)}|^{2}}{q_{k,n}(\boldsymbol{W}^{(t)},\boldsymbol{v}^{(t)})}) \\
&\triangleq f_{k,n}^{\mathrm{E}}(\boldsymbol{W},\boldsymbol{v};\boldsymbol{W}^{(t)},\boldsymbol{v}^{(t)}), \quad (21)
\end{aligned}$$

where

$$q_{k,n}(\boldsymbol{W},\boldsymbol{\nu};\boldsymbol{W}^{(t)},\boldsymbol{\nu}^{(t)}) = \delta_n^2 + \sum_{i\neq k}^K \Re(\boldsymbol{w}_i^{(t)})^H \boldsymbol{G}_n^H \boldsymbol{\nu}^{(t)} \times (2\boldsymbol{\nu}^H \boldsymbol{G}_n \boldsymbol{w}_i - (\boldsymbol{\nu}^H)^{(t)} \boldsymbol{G}_n \boldsymbol{w}_i^{(t)})\}, \quad (22)$$

Proof: Please refer to Appendix A.

Then, from (20) and (21), we know the lower bound of $\mathcal{R}(W, v)$ is given by

$$\mathcal{R}(\boldsymbol{W}, \boldsymbol{\nu}) = \min_{\forall k, \forall n} \left\{ f_k^{\mathrm{B}}(\boldsymbol{W}, \boldsymbol{\nu}) - f_{k,n}^{\mathrm{E}}(\boldsymbol{W}, \boldsymbol{\nu}) \right\}$$

$$\geq \min_{\forall k, \forall n} \left\{ f_k^{\mathrm{B}}(\boldsymbol{W}, \boldsymbol{\nu}; \boldsymbol{W}^{(t)}, \boldsymbol{\nu}^{(t)}) - f_{k,n}^{\mathrm{E}}(\boldsymbol{W}, \boldsymbol{\nu}; \boldsymbol{W}^{(t)}, \boldsymbol{\nu}^{(t)}) \right\}$$

$$\stackrel{\Delta}{=} \mathcal{R}^{\mathrm{lb}}(\boldsymbol{W}, \boldsymbol{\nu}; \boldsymbol{W}^{(t)}, \boldsymbol{\nu}^{(t)}).$$
(23)

Note that, according to (20) and (21), the equality in (23) holds when $W = W^{(t)}$ and $v = v^{(t)}$.

Thus, a family of the approximated problems of (P2) is given as follows:

$$(\mathbf{P}2-t) : \max_{\mathbf{W}, \mathbf{v}} \mathcal{R}^{\text{lb}}(\mathbf{W}, \mathbf{v}; \mathbf{W}^{(t)}, \mathbf{v}^{(t)})$$

s.t. (14a) and (19a).

However, $(\mathbf{P}2 - t)$ is still a non-convex problem due to the following reasons:

- First, W and v are coupled in the terms of $v^H H_k w_k$ and $v^H G_n w_k$, which makes the objective function $\mathcal{R}^{\text{lb}}(W, v; W^{(t)}, v^{(t)})$ not jointly concave with respect to (W, v).
- Second, it is straightforward to know that (19a) with $\Phi = \Phi_1$ is a convex set but a non-convex set with $\Phi = \Phi_2$ and $\Phi = \Phi_3$.

In subsection IV-B, we will first develop alternating optimization method to deal with the coupled optimization variables in ($\mathbf{P}2-t$) with $\mathbf{\Phi} = \mathbf{\Phi}_1$, and then we will extend it to the scenarios with $\mathbf{\Phi} = \mathbf{\Phi}_2$ and $\mathbf{\Phi} = \mathbf{\Phi}_3$, respectively.

B. ALTERNATING OPTIMIZATION WITH CONTINUOUS AND DISCRETE REFLECTING COEFFICIENTS

1) THE SOLUTION OF (P2) WITH $\Phi=\Phi_1$

In this part, we develop the alternating optimization to solve $(\mathbf{P}2 - t)$ when $\mathbf{\Phi} = \mathbf{\Phi}_1$ in constraint (19a), which leads constraint (19a) to be a convex set. Hence, the non-convexity of $(\mathbf{P}2 - t)$ only stems from the coupled optimization variables.

In fact, although the objective function $\mathcal{R}^{\text{lb}}(W, v; W^{(t)}, v^{(t)})$ is non-concave due to the coupled W and $v, f_k^{\text{B}}(W, v; W^{(t)}, v^{(t)})$ in (20) is biconcave in W and v, i.e., $f_k^{\text{B}}(W, v; W^{(t)}, v^{(t)})$ is concave both in W with fixed v and in v with fixed W. Similarly, for the domain $q_{k,n}(W, v; W^{(t)}, v^{(t)}) \geq 0$, the function $\frac{|v^H G_n w_k|^2}{q_{k,n}(W, v; W^{(t)}, v^{(t)})}$ in (21) is a biconvex function with respect to W and v, which leads to a biconvex function $f_{k,n}^{\text{E}}(W, v; W^{(t)}, v^{(t)})$ with respect to W and v. Hence, $\mathcal{R}^{\text{lb}}(W, v; W^{(t)}, v^{(t)})$ is a biconcave function in W and v.

Therefore, we know $(\mathbf{P}2 - t)$ with $\mathbf{\Phi} = \mathbf{\Phi}_1$ has convex constraints and concave objective function in W with fixed v and in v with fixed W. Hence, we can apply the alternating optimization method to solve $(\mathbf{P}2-t)$ in an alternating manner efficiently. Specifically, the alternating algorithm decouples $(\mathbf{P}2-t)$ into the following two subproblems for the optimization of W and v, respectively,

$$(\mathbf{P3}-\mathbf{A}): \max_{\mathbf{W}} \mathcal{R}^{lb}(\mathbf{W}, \mathbf{v}; \mathbf{W}^{(t)}, \mathbf{v}^{(t)})$$

s.t. (14a),

and

$$(\mathbf{P}3-\mathbf{B}) : \max_{\boldsymbol{\nu}} \mathcal{R}^{\mathrm{lb}}(\boldsymbol{W}, \boldsymbol{\nu}; \boldsymbol{W}^{(t)}, \boldsymbol{\nu}^{(t)})$$

s.t. (19a) with $\boldsymbol{\Phi} = \boldsymbol{\Phi}_1$.

Note that $(\mathbf{P3} - \mathbf{A})$ is an optimization subproblem for solving W with a given v and $(\mathbf{P3} - \mathbf{B})$ is an optimization subproblem for solving v with a given W.

As aforementioned, we know both $(\mathbf{P}3 - \mathbf{A})$ and $(\mathbf{P}3 - \mathbf{B})$ are convex optimization problems, which can be solved optimally and efficiently by using CVX [35]. Thus, problem $(\mathbf{P}2)$ with $\mathbf{\Phi} = \mathbf{\Phi}_1$ can be solved efficiently by alternately solving $(\mathbf{P}3 - \mathbf{A})$ and $(\mathbf{P}3 - \mathbf{B})$ in an iterative manner of path-following algorithm. In particular, the algorithm steps of the alternating optimization based path-following algorithm are summarized in Algorithm 1.

Algorithm 1 Alternating Optimization Based Path-Following Algorithm

- 1: Initialize $W^{(0)}$, $v^{(0)}$ and t = 0.
- 2: repeat
- 3: $t \leftarrow t+1$,
- 4: Set $v = v^{(t-1)}$ and calculate $W^{(t)}$ by solving the convex optimization problem (P3 A),
- 5: Set $W = W^{(t)}$ and calculate $v^{(t)}$ by solving the convex optimization problem (P3 B),

6: **until**
$$\Gamma = \frac{\left(\mathcal{R}(W^{(t)}, \boldsymbol{\nu}^{(t)}) - \mathcal{R}(W^{(t-1)}, \boldsymbol{\nu}^{(t-1)})\right)}{\mathcal{R}(W^{(t)}, \boldsymbol{\nu}^{(t)})}$$
 converges.

2) THE SOLUTIONS OF (P2) WITH $\Phi = \Phi_2$

In this part, we extend the above alternating optimization to solve $(\mathbf{P}2 - t)$ when $\mathbf{\Phi} = \mathbf{\Phi}_2$ in constraint (19a), which leads constraint (19a) to be a non-convex set. To handle this non-convex constraint, we propose the following two methods:

• In the first method, we introduce a positive constant relaxation factor λ to reformulate ($\mathbf{P}2 - t$) with $\mathbf{\Phi} = \mathbf{\Phi}_2$ as the following optimization problem,

$$(\mathbf{P}4-t) : \max_{\mathbf{W}, \mathbf{v}} \mathcal{R}^{\text{lb}}(\mathbf{W}, \mathbf{v}; \mathbf{W}^{(t)}, \mathbf{v}^{(t)}) + \lambda \sum_{l=1}^{L+1} |v_l|^2$$

s.t. (14a) and (19a) with $\mathbf{\Phi} = \mathbf{\Phi}_1$.

Note that the added nonnegative quadratic term $\lambda \sum_{l=1}^{L+1} |v_l|^2$ attempts to force the inequality holds for v_l , i.e., $|v|_l = 1$.

However, the objective of $(\mathbf{P4} - t)$ is to maximize the summation of concave and convex functions, which belongs to a non-convex problem. To further deal with this challenge, we use the first-order Taylor series expansion to approximates the convex function as an affine function [36]. Then, we iteratively solve the approximated convex optimization problem until the convergence is met. Specifically, the approximated

problem is

$$(\mathbf{P}4 - \mathbf{A}) : \max_{\boldsymbol{W}, \boldsymbol{v}} \mathcal{R}^{\mathrm{lb}}(\boldsymbol{W}, \boldsymbol{v}; \boldsymbol{W}^{(t)}, \boldsymbol{v}^{(t)}) \\ + \lambda \sum_{l=1}^{L+1} \Re \left\{ (v_l^{(t)})^H (2v_l - v_l^{(t)}) \right\}$$

s.t. (14a), and (19a) with $\boldsymbol{\Phi} = \boldsymbol{\Phi}_1$.

Finally, the only non-convex term in ($\mathbf{P}4 - \mathbf{A}$) stems from the coupled W and v in the objective function $\mathcal{R}^{\text{lb}}(W, v; W^{(i)}, v^{(i)})$, which can be solved efficiently by applying the same alternating optimization method, i.e., Algorithm 1.

However, this method has the main drawbacks that we use the approximation in $(\mathbf{P4} - t)$ and there is no beforehand choice for the relaxation factor λ to speed up the convergence [32]. Hence, we further propose the direct projection method in the next part.

In the second method, we can apply the projection method to project the solution of (P2) with $\Phi = \Phi_1$ into $\Phi = \Phi_2$ directly. Specifically, denote the solutions of (P2) with $\Phi = \Phi_1$ and (P2) with $\Phi = \Phi_2$ as $(W^{\dagger}, v^{\dagger})$ and $(W^{\ddagger}, v^{\ddagger})$, respectively. Thus, the $(W^{\ddagger}, v^{\ddagger})$ can be obtained by solving the following projection problem:

$$(\mathbf{P}4 - \mathbf{B}): \mathbf{v}^{\ddagger} = \underset{\mathbf{v}}{\operatorname{arg\,min}} \left\| \mathbf{v} - \mathbf{v}^{\dagger} \right\|^{2}$$

s.t. (19a) with $\mathbf{\Phi} = \mathbf{\Phi}_{2}$.

From [23], the optimal solution to $(\mathbf{P4} - \mathbf{B})$ is given by

$$\mathbf{v}^{\ddagger} = \exp(j \arg(\mathbf{v}^{\dagger})), \qquad (27)$$

and $W^{\ddagger} = W^{\dagger}$. Note that, to further improve the performance of this method, we can rerun the following adjusted Algorithm 1 to iteratively update the obtained $(W^{\ddagger}, v^{\ddagger})$:

- Initialize $W^0 = W^{\ddagger}$, $v^0 = v^{\ddagger}$ and t = 0. Then, perform the following steps iteratively until the objective function converges.
- $t \leftarrow t + 1$, set $v = v^{(t-1)}$ and calculate $W^{(t)}$ by solving the convex optimization problem (P3 A),
- Set $W = W^{(t)}$, project the solution of (P3 B) into Φ_2 by (27), and denote it as \tilde{v} ,
- Update $v^{(t)}$ using the following rule:

$$\boldsymbol{\nu}^{(t)} = \begin{cases} \tilde{\boldsymbol{\nu}}, & \text{if } \mathcal{R}(\boldsymbol{W}^{(t)}, \tilde{\boldsymbol{\nu}}) \geq \mathcal{R}(\boldsymbol{W}^{(t)}, \boldsymbol{\nu}^{(t-1)}), \\ \boldsymbol{\nu}^{(t-1)}, & \text{otherwise.} \end{cases}$$
(28)

3) THE SOLUTION OF (P2) WITH $\Phi = \Phi_3$

In this part, we develop algorithms to solve (**P**2) when $\Phi = \Phi_3$ in constraint (19a), which leads the optimized problem belongs to a class of combinatorial optimization problem, which is an NP-hard problem in general. Thus, it will cause intractable complexity to obtain the optimal solution. Hence, we will use the similar heuristic projection method in the above to solve this problem efficiently.

To begin with, we denote the solution of $\Phi = \Phi_3$ as (W^{\S}, v^{\S}) . Then, we can directly project v^{\dagger} , the solution of **(P2)** with $\Phi = \Phi_1$, into Φ_3 to obtain (W^{\S}, v^{\S}) , i.e.,

$$v_l^{\S} = \begin{cases} e^{j\varphi_{\hat{q}}}, & \text{where } \hat{q} = \underset{1 \le q \le Q}{\arg\min} \left| v_l^{\dagger} - e^{j\varphi_q} \right|, 1 \le l \le L, \\ 1, & l = L + 1, \end{cases}$$

$$(29)$$

where v_l^{\S} and v_l^{\dagger} are the *l*-th element of v^{\S} and v^{\dagger} , respectively. $W^{\S} = W^{\dagger}$. Note that the rest steps to update (W^{\S}, v^{\S}) are similar as the second method in subsection IV-B.2, which is omitted here for brevity.

C. CONVERGENCE ANALYSIS

In this part, we analyze the convergence of the proposed alternating optimization based path-following algorithm, i.e., Algorithm 1, which is given in the following theorem.

Theorem 1: The value of the objective function increases in each iteration of Algorithm 1, i.e., $\mathcal{R}(\mathbf{W}^{(t)}, \mathbf{v}^{(t)}) \leq \mathcal{R}(\mathbf{W}^{(t+1)}, \mathbf{v}^{(t+1)})$, which guarantees to converge to a local (global) optimum.

Proof: To begin with, we have

$$\mathcal{R}(\boldsymbol{W}^{(t)}, \boldsymbol{v}^{(t)}) = \mathcal{R}^{\text{lb}}(\boldsymbol{W}^{(t)}, \boldsymbol{v}^{(t)}; \boldsymbol{W}^{(t)}, \boldsymbol{v}^{(t)})$$

$$\leq \left\{ \max_{\boldsymbol{W}} \mathcal{R}^{\text{lb}}(\boldsymbol{W}, \boldsymbol{v}^{(t)}; \boldsymbol{W}^{(t)}, \boldsymbol{v}^{(t)}) \right\}$$

$$= \mathcal{R}^{\text{lb}}(\boldsymbol{W}^{(t+1)}, \boldsymbol{v}^{(t)}; \boldsymbol{W}^{(t)}, \boldsymbol{v}^{(t)})$$

$$\leq \left\{ \max_{\boldsymbol{v}} \mathcal{R}^{\text{lb}}(\boldsymbol{W}^{(t+1)}, \boldsymbol{v}; \boldsymbol{W}^{(t)}, \boldsymbol{v}^{(t)}) \right\}$$

$$= \mathcal{R}^{\text{lb}}(\boldsymbol{W}^{(t+1)}, \boldsymbol{v}^{(t+1)}; \boldsymbol{W}^{(t)}, \boldsymbol{v}^{(t)})$$

$$\leq \mathcal{R}(\boldsymbol{W}^{(t+1)}, \boldsymbol{v}^{(t+1)}), \qquad (30)$$

where (a) is because the equality in (23) holds when $W = W^{(t)}$ and $v = v^{(t)}$, (b) and (c) are because $W^{(t+1)}$ and $v^{(t+1)}$ are the optimal solutions of the convex optimization problems of (**P**3 – A) and (**P**3 – B), respectively, and (d) is because $\mathcal{R}^{\text{lb}}(W^{(t+1)}, v^{(t+1)}; W^{(t)}, v^{(t)})$ is the lower bound of the function $\mathcal{R}(W, v)$ in (23).

Furthermore, due to (14a) and (19a), we know $W^{(t)}$ and $v^{(t)}$ are both bounded. According to Cauchy's theorem [32], we know the sequence of $(W^{(t)}, v^{(t)})$ will converge to (W^*, v^*) as $t \to \infty$, i.e.,

$$0 = \lim_{t \to \infty} \left\{ \mathcal{R}(\boldsymbol{W}^{(t)}, \boldsymbol{\nu}^{(t)}) - \mathcal{R}(\boldsymbol{W}^*, \boldsymbol{\nu}^*) \right\}$$

$$\leq \lim_{t \to \infty} \left\{ \mathcal{R}(\boldsymbol{W}^{(t+1)}, \boldsymbol{\nu}^{(t+1)}) - \mathcal{R}(\boldsymbol{W}^*, \boldsymbol{\nu}^*) \right\} = 0.$$
(31)

Hence, we have proved the $\mathcal{R}(\mathbf{W}^{(t)}, \mathbf{v}^{(t)}) \leq \mathcal{R}(\mathbf{W}^{(t+1)}, \mathbf{v}^{(t+1)})$, which can guarantee to converge to a local (global) optimum.

Theorem 2: The corresponding solution (W^*, v^*) will converge to a *Karush-Kuhn-Tucker* (KKT) point finally.

Proof: Please refer to Appendix B

However, although the convergence of the proposed algorithm is guaranteed, it requires solving the convex optimization problems (**P**3 – A) and (**P**3 – B) whose complexity are in the order of $\mathcal{O}((KN + 1)K^2M^2)$ and $\mathcal{O}((KN + L + 1)(L + 1)^2)$ [37], respectively. Hence, we will develop the methods to reduce the computational complexity in the next subsection. Note that, since the solutions with Φ_1 can be projected into Φ_2 and Φ_3 directly, thus we only consider $\Phi = \Phi_1$ when developing the low-complexity algorithms.

V. SUBOPTIMAL ALGORITHMS WITH LOW-COMPLEXITY

In this section, two suboptimal algorithms are developed to further reduce the complexity. Firstly, we develop an alternating optimization algorithm for the case with one legitimate user and one eavesdropper, where the closed-form solutions are provided in each iteration. Then, we develop an noniterative suboptimal algorithm based on ZF beamforming for the case with multiple legitimate users and eavesdroppers.

A. ALTERNATING OPTIMIZATION FOR (P2) WITH K = 1 AND N = 1

In this section, we develop a low-complexity algorithm to solve (**P**2) for the case with K = 1 and N = 1. Although there is no-interference in the objective function, the problem is still non-convex and hard to solve due to the coupled W and v. Thus, we also need to apply alternating optimization to decouple W and v. Fortunately, we can obtain the closed-form solutions in each iteration, which leads it to be a low-complexity algorithm.

Specifically, (P2) with K = 1 and N = 1 can be solved by alternately solving the following two subproblems:

$$(\mathbf{P}5-A) : \max_{w_1} \ln\left(1 + |\tilde{\boldsymbol{h}}_1^H w_1|^2\right) - \ln\left(1 + |\tilde{\boldsymbol{g}}_1^H w_1|^2\right)$$

s.t. $\|w_1\|^2 \le P$, (32)

which is an optimization problem of w_1 for a given v with $\tilde{h}_1^H = v^H H_1$ and $\tilde{g}_1^H = v^H G_1$, and

$$(\mathbf{P5}-\mathbf{B}): \max_{\mathbf{v}} \ln\left(1 + \left|\mathbf{v}^{H}\bar{\mathbf{h}}_{1}\right|^{2}\right) - \ln\left(1 + \left|\mathbf{v}^{H}\bar{\mathbf{g}}_{1}\right|^{2}\right)$$

s.t. $|v_{l}|^{2} \le 1, \quad 1 \le l \le L+1,$ (33)

which is an optimization problem of v for a given w_1 with $\bar{h}_1 = H_1 w_1$ and $\bar{g}_1 = G_1 w_1$. Note that in constraint (33), we have relaxed constraint (19a) due to v_{L+1} . To make the solution of v in the above problem (P5 – B) to satisfy (19a), we need to set v_{L+1}^* as 1 and $v_l^* = v_l / \exp(j \arg(v_{L+1}))$ for $1 \le l \le L$ after the convergence. Besides, constraints (33) can be regarded as the per-antenna power constraints with the maximum power of one [38].

In the following two parts, we provide the solutions to (P5 - A) and (P5 - B), respectively.

1) THE OPTIMAL SOLUTION TO (P5 - A)

This problem is the downlink MISO beamforming problem for basic wiretap channel, which has been studied in [39]. In particular, the optimal solution to $(\mathbf{P5} - \mathbf{A})$ is given by

$$\boldsymbol{w}_{1}^{*} = \sqrt{P} \frac{\left(\boldsymbol{I} + P\tilde{\boldsymbol{g}}_{1}\tilde{\boldsymbol{g}}_{1}^{H}\right)^{-\frac{1}{2}}\boldsymbol{q}}{\left\|\left(\boldsymbol{I} + P\tilde{\boldsymbol{g}}_{1}\tilde{\boldsymbol{g}}_{1}^{H}\right)^{-\frac{1}{2}}\boldsymbol{q}\right\|},$$
(34)

where q is the eigenvector of matrix Z corresponding to the largest eigenvalue, and

$$\boldsymbol{Z} = (\boldsymbol{I} + P\tilde{\boldsymbol{g}}_{1}\tilde{\boldsymbol{g}}_{1}^{H})^{-\frac{1}{2}}(\boldsymbol{I} + P\tilde{\boldsymbol{h}}_{1}\tilde{\boldsymbol{h}}_{1}^{H})(\boldsymbol{I} + P\tilde{\boldsymbol{g}}_{1}\tilde{\boldsymbol{g}}_{1}^{H})^{-\frac{1}{2}}.$$
 (35)

2) THE SOLUTIONS TO (P5 - B)

In this part, we develop the efficient algorithm to solve (P5 - B) in an iterative manner,² and we provide the closed-form solutions in each iteration. To do so, we first introduce the following lemma.

Lemma 2: Let x be a positive real number, and define $f(y) = -xy + \ln y + 1$, then we have $-\ln x = \max_{y>0} f(y)$, and the optimal corresponding solution in the right hand side of this equation is $y^* = 1/x$.

Proof: This proof is similar in [41], which is omitted here for brevity.

According to lemma 2, $(\mathbf{P5} - \mathbf{B})$ can be equivalently rewritten as

$$(\mathbf{P6}): \max_{\mathbf{v}, y} \ln(1 + \left| \mathbf{v}^{H} \bar{\mathbf{h}}_{1} \right|^{2}) - y(1 + \left| \mathbf{v}^{H} \bar{\mathbf{g}}_{1} \right|^{2}) + \ln y$$

s.t. $y > 0$ and (33). (36)

This reformulated problem is still non-convex, but it is convex in v or y with the other variable is fixed. Then, we can further apply alternating optimizing to optimize v and y in an iterative manner. Specifically, the alternating optimization subproblems are given as follows:

$$(\mathbf{P}6-\mathbf{A}): \max_{y>0} -y(1+\left|\mathbf{v}^{H}\bar{\mathbf{g}}_{1}\right|^{2})+\ln y,$$

which is a convex optimization problem of y for a given v, and

$$(\mathbf{P}6-\mathbf{B}): \max_{V \succeq 0} \ln(1 + \bar{\boldsymbol{h}}_1^H \boldsymbol{V} \bar{\boldsymbol{h}}_1) - y \bar{\boldsymbol{g}}_1^H \boldsymbol{V} \bar{\boldsymbol{g}}_1$$

s.t. $\operatorname{Tr}(\boldsymbol{e}_l^H \boldsymbol{V} \boldsymbol{e}_l) \le 1, \quad 1 \le l \le L+1, \quad (37)$

which is a convex optimization problem of $V = \nu \nu^H$ for a given y, where $e_l \in \mathbb{R}^{(L+1)\times 1}$ is a unit vector with the *l*-th entry being one and other entries being zero. Note that since the rank of the optimal V of (P6 – B) must be one, which will be proved in Appendix C, there is no need to add the rank-one constraint for variable V.

In the following, we show the optimal solutions to (P6-A) and (P6-B), respectively. Specifically, according to

²Note that although (**P**5–*B*) can be solved by transforming it into a convex semidefinite programming (SDP) by applying semidefinite relaxation (SDR) technique, but solving the SDP problems requires very high complexity, which is in the order of $\mathcal{O}((N + 1)^{4.5})$ [40].

Lemma 2, it is straightforward to know the optimal solution to $(\mathbf{P}6 - \mathbf{A})$ is given by

$$y^* = (1 + \left| \boldsymbol{v}^H \bar{\boldsymbol{g}}_1 \right|^2)^{-1},$$
 (38)

Next, since (P6 - B) is a convex problem, it is straightforward to know that strong duality holds for (P6 - B) [42]. Hence, we can obtain its optimal solution by solving its dual problem. To begin with, the Lagrangian dual of problem (P6 - B) can be written as

$$(\mathbf{P}6-\mathbf{C}):\min_{\boldsymbol{\chi}} \mathcal{L}(\boldsymbol{\chi}) \quad \text{s.t.} \ \boldsymbol{\chi}_l \ge 0, \quad 1 \le l \le L+1, \quad (39)$$

where $\boldsymbol{\chi} = [\chi_1, \cdots, \chi_{L+1}]^T \in \mathbb{R}^{(L+1) \times 1}$ is the dual variable and

$$\mathcal{L}(\boldsymbol{\chi}) = \max_{\boldsymbol{V} \succeq 0} \left\{ \ln(1 + \bar{\boldsymbol{h}}_{1}^{H} \boldsymbol{V} \bar{\boldsymbol{h}}_{1}) - y \bar{\boldsymbol{g}}_{1}^{H} \boldsymbol{V} \bar{\boldsymbol{g}}_{1} + \sum_{l=1}^{L+1} \chi_{l} (1 - \boldsymbol{e}_{l}^{H} \boldsymbol{V} \boldsymbol{e}_{l}) \right\}.$$
(40)

Then, the optimal solution to $(\mathbf{P6} - \mathbf{B})$ can be obtained by iteratively solving (40) with fixed χ and updating χ by subgradient methods, e.g., the ellipsoid method. The details for the subgradient methods have been studied in [43], which are omitted here for brevity. Then, we have the following theorem.

Theorem 3: If the optimal χ^* is given, the optimal solution for $(\mathbf{P}6 - \mathbf{B})$ is given by $V^* = v^*(v^*)^H$ with

$$\boldsymbol{\nu}^* = \frac{\sqrt{\left(1 - \frac{1}{(\varepsilon^*)^2}\right)^+}}{\varepsilon^*} \left(\operatorname{diag}\left(\boldsymbol{\chi}^*\right) + y \bar{\boldsymbol{g}}_1 \bar{\boldsymbol{g}}_1^H\right)^{-1} \bar{\boldsymbol{h}}_1, \quad (41)$$

where $\varepsilon^* = \left\| \left(\operatorname{diag}(\boldsymbol{\chi}^*) + y \bar{\boldsymbol{g}}_1 \bar{\boldsymbol{g}}_1^H \right)^{-0.5} \bar{\boldsymbol{h}} \right\|.$

Proof: Please refer to Appendix C.

The convergence for the above algorithm is also guaranteed, the proof is similar to the proof of theorem 1, which is omitted here for brevity. Besides, the detailed steps of the above algorithm are summarized in Algorithm 2.

Algorithm 2 Alternating Optimization Method	
1:	Initialize $w_1^{(0)}$, $v^{(0)}$, $y^{(0)}$ and $t = 0$.
2:	repeat
3:	$t \leftarrow t+1,$
4:	Set $\mathbf{v} = \mathbf{v}^{(t-1)}$, then calculate $\mathbf{W}^{(t)}$ by (34),
5.	Set $\mathbf{v} = \mathbf{v}^{(t-1)}$ and $\mathbf{w}_1 = \mathbf{w}^{(t)}$ then calculate

- 5: Set $\mathbf{v} = \mathbf{v}^{(t-1)}$ and $\mathbf{w}_1 = \mathbf{w}_1^{(t)}$, then calculate $y^{(t)}$ by (38),
- 6: Initialize $\boldsymbol{\chi} = \boldsymbol{\chi}^{(0)}$ and i = 0, set $\boldsymbol{w}_1 = \boldsymbol{w}_1^{(t)}$ and $y = y^{(t)}$,
- 7: repeat
- 8: $i \leftarrow i+1$, set $\chi = \chi^{(i-1)}$, then calculate $\nu^{(t)}$ by (41),
- 9: Calculate $\chi^{(i)}$ according to the ellipsoid method [43],
- 10: **until** Convergence.
- 11: **until** Convergence.

B. HEURISTIC ALGORITHM FOR (P2) WITH $K \ge 1$ AND $N \ge 1$

In this subsection, we provide the heuristic algorithm for (**P**2) with $K \ge 1$ and $N \ge 1$. Specifically, when $L \to \infty$, we can assume the received signal from the BS to users can be ignored due to the total powers of received signals are dominated by the signals from the BS and through the IRS to the users with asymptotically large *L*. In addition, according to the Rician channel model introduced in (1) and (4), when $\kappa_F \to \infty$ and $\kappa_{h_{r,k}} \to \infty$ and $\kappa_{g_{r,n}} \to \infty$, the channel responses from the BS to the IRS, from the IRS to B_k and from the IRS to E_n are dominated by LoS components since the NLoS components can be practically ignored. Besides, we also assume all legitimate receivers are in the same directions to the IRS, i.e., $\vartheta_{r,k} = \vartheta_{r,i}$ holds for $1 \le i, k \le K$. Hence, the total power of received signals at the legitimate receivers can be given by

$$\left|\boldsymbol{a}_{L}^{H}\left(\vartheta_{r,k}\right)\boldsymbol{\Theta}\boldsymbol{a}_{L}(\vartheta^{\text{AoA}})\right|^{2}\left\|\left(\boldsymbol{a}_{M}^{H}(\vartheta^{\text{AoD}})\right)\sum_{k=1}^{K}\boldsymbol{w}_{k}\right\|^{2},\qquad(42)$$

Then, it is straightforward to know the optimal θ^* to maximize the total received signal power in (42) is given by

$$\theta_l^* = \exp\left(j\frac{(l-1)2\pi\lambda}{d}(\sin\vartheta_{r,k} - \sin\vartheta^{\text{AoA}})\right), \ 1 \le l \le L.$$
(43)

Substituting (43) into (11) and (12), the throughput of *k*-th confidential message at B_k and E_n can be written as

$$\hat{R}_{k}^{\mathrm{B}} = \ln\left(1 + \frac{\left|\hat{\boldsymbol{h}}_{k}^{H}\boldsymbol{w}_{k}\right|^{2}}{\sum_{i\neq k}^{K}\left|\hat{\boldsymbol{h}}_{k}^{H}\boldsymbol{w}_{i}\right|^{2} + \sigma_{k}^{2}}\right), \qquad (44)$$

$$\hat{R}_{k,n}^{\mathrm{E}} = \ln\left(1 + \frac{\left|\hat{\boldsymbol{g}}_{n}^{H}\boldsymbol{w}_{k}\right|^{2}}{\sum_{i\neq k}^{K}\left|\hat{\boldsymbol{g}}_{n}^{H}\boldsymbol{w}_{i}\right|^{2} + \delta_{n}^{2}}\right),\qquad(45)$$

respectively, where $\hat{\boldsymbol{h}}_{k}^{H} = (\boldsymbol{h}_{r,k}^{H}\boldsymbol{\Theta}\boldsymbol{F} + \boldsymbol{h}_{d,k}^{H})$ and $\hat{\boldsymbol{g}}_{n}^{H} = (\boldsymbol{g}_{r,n}^{H}\boldsymbol{\Theta}\boldsymbol{F} + \boldsymbol{g}_{d,n}^{H})$. Next, we apply the ZF beamforming scheme,³ which forces the information leakage to E_{n} to be zero, i.e.,

$$\hat{\boldsymbol{G}}^{H}\boldsymbol{W}=\boldsymbol{0}, \tag{46}$$

where $\hat{\boldsymbol{G}} = [\hat{\boldsymbol{g}}_1, \cdots, \hat{\boldsymbol{g}}_N] \in \mathbb{C}^{M \times N}$. Thus, the ZF beamformer \boldsymbol{W} can be expressed as

$$W = XUP, \tag{47}$$

where $X \in \mathbb{C}^{M \times (M-N)}$ consists of (M - N) singular vectors of \hat{G} corresponding to the zero singular values, $U = [u_1, \cdots, u_K] \in \mathbb{C}^{(M-N) \times K}$ subject to $||u_k|| = 1$ for $1 \le k \le K \cdot P = \text{diag}(p) \in \mathbb{R}^{K \times K}$ with $p = [p_1, \cdots, p_K]^T \in \mathbb{R}^{K \times 1}$ subject to $\sum_{k=1}^{K} p_k = P$.

³Note that $M \ge N$ is required in the studied system if ZF beamforming is adopted.

Therefore, problem (**P**2) can be changed as the following the minimal-SINR maximization problem, i.e.,

$$(\mathbf{P7}): \max_{\boldsymbol{U},\boldsymbol{p}} \min_{\forall k} \gamma_k \stackrel{\Delta}{=} \frac{p_k \left| \hat{\boldsymbol{x}}_k^H \boldsymbol{u}_k \right|^2}{\sum_{i \neq k}^K p_i \left| \hat{\boldsymbol{x}}_k^H \boldsymbol{u}_i \right|^2 + 1}$$

s.t. $\sum_{k=1}^K p_k = P$ and $\|\boldsymbol{u}_k\| = 1$.

where $\hat{\mathbf{x}}_{k}^{H} = \hat{\mathbf{h}}_{k}^{H} X / \sigma_{k}$. From [44], we know the optimal beamformer in problem (**P**2) has the following structure

$$\boldsymbol{u}_{k}^{*} = \frac{\left(\sum_{i\neq k}^{K} z_{i} \hat{\boldsymbol{x}}_{i} \hat{\boldsymbol{x}}_{i}^{H} + \boldsymbol{I}_{K}\right)^{-1} \hat{\boldsymbol{x}}_{k}}{\left\| \left(\sum_{i\neq k}^{K} z_{i} \hat{\boldsymbol{x}}_{i} \hat{\boldsymbol{x}}_{i}^{H} + \boldsymbol{I}_{K}\right)^{-1} \hat{\boldsymbol{x}}_{k} \right\|},$$
(49)

where the unique and positive z_k^4 for $1 \le k \le K$ can be obtained by solving the following equations [44]:

$$z_k^* = \frac{\gamma^*}{\hat{\mathbf{x}}_k^H \left(\sum_{i \neq k}^K \hat{\mathbf{x}}_i \hat{\mathbf{x}}_i^H + \mathbf{I}_K\right)^{-1} \hat{\mathbf{x}}_k},\tag{50}$$

$$\gamma^{*} = \frac{P}{\sum_{k=1}^{K} \left(\hat{\mathbf{x}}_{k}^{H} (\sum_{i \neq k}^{K} \hat{\mathbf{x}}_{i} \hat{\mathbf{x}}_{i}^{H} + \mathbf{I}_{K})^{-1} \hat{\mathbf{x}}_{k} \right)^{-1}}, \qquad (51)$$

Note that γ^* is the optimal value of the objective function in (**P**7), and we know $\gamma_k^* = \gamma^*$ holds for all $1 \le k \le K$ [45]. Using this fact, we have the following equation:

$$\frac{\boldsymbol{p}^*}{\boldsymbol{\gamma}^*} = \boldsymbol{\Xi} \boldsymbol{Y} \boldsymbol{p}^* + \boldsymbol{\Xi} \boldsymbol{1}_K, \tag{52}$$

where $\Xi = \text{diag}([\frac{1}{|\hat{\mathbf{x}}_{i}^{H}\boldsymbol{u}_{i}|^{2}}, \cdots, \frac{1}{|\hat{\mathbf{x}}_{k}^{H}\boldsymbol{u}_{k}|^{2}}]) \in \mathbb{R}^{K \times K}$, $Y = [Y_{ik}] \in \mathbb{R}^{K \times K}$ with $Y_{ik} = |\hat{\mathbf{x}}_{i}^{H}\boldsymbol{u}_{k}|^{2}$ if $i \neq k$ and $Y_{ik} = 0$ if i = k. Then, we know the optimal power allocation p is given as follows

$$\boldsymbol{p}^* = \boldsymbol{\gamma}^* \left(\boldsymbol{I}_k - \boldsymbol{\gamma}^* \boldsymbol{\Xi} \boldsymbol{Y} \right) \boldsymbol{\Xi} \boldsymbol{1}_K.$$
 (53)

VI. SIMULATION RESULTS

In this section, we present simulation results to validate the advantages of using the IRS to improve the secret communication of the downlink MISO broadcast system. It is assumed that the noise variances σ_k^2 at B_k and δ_n^2 at E_n are the same and normalized to one. The maximum transmit power *P* is defined in dB with respect to the noise variance. For performance comparison, we also show the performances of two suboptimal baseline schemes. In particular, for the "Rand" baseline, we randomly select the reflecting coefficients of the IRS from Φ_2 with equal probability and apply Algorithm 1 without updating the reflecting coefficients anymore to obtain the beamforming design; and for the "Without IRS" baseline, we assume that the channels from the BS to the IRS, from the IRS to B_k , and from the IRS to E_n are blocked, i.e., F = 0, $h_{r,k}^H = 0$, and $g_{r,n}^H = 0$, respectively, with $1 \le k \le K$ and

⁴In fact, z_k is the power allocation for the virtual dual uplink network [44], which is strictly positive

-7,1

 $1 \leq n \leq N$, and then we apply Algorithm 1 to obtain the beamforming design. This represents the worst-case scenario to achieve secret communication in the absence of the IRS. In addition, we assume that the legitimate receivers and the eavesdroppers are in the same directions to the BS and the IRS, i.e., $\vartheta_{d,k} = \vartheta_{d,i}$ and $\vartheta_{r,k} = \vartheta_{r,i}$ for $1 \leq i, k \leq K$, and $\tilde{\vartheta}_{d,n} = \tilde{\vartheta}_{d,i}$ and $\tilde{\vartheta}_{r,n} = \tilde{\vartheta}_{r,i}$ for $1 \leq i, n \leq N$. We also assume that $\vartheta_{r,k}, \tilde{\vartheta}_{r,n}, \vartheta^{AoA}$ and ϑ^{AoD} are uniformly distributed between $[0, 2\pi)$, while $\vartheta_{d,k}$ and $\tilde{\vartheta}_{d,n}$ are uniformly distributed between $[-\pi/3, \pi/3]$.



FIGURE 2. Minimum secrecy rate versus the maximum transmit power of the BS: M = L = 5, and $\kappa_h = 1$ with $h \in \mathcal{H}$: (a) K = N = 2; (b) K = N = 1.

Figure 2(a) and Fig. 2(b) show the minimum secrecy rate under different values of the maximum transmit power P of the BS for K = N = 2 and K = N = 1, respectively. Note that "Algorithm 1: $\Phi_2(1)$ " and "Algorithm 1: $\Phi_2(2)$ " denote Algorithm 1 with the first method based on Taylor series expansion and second method based on projection for $\Phi = \Phi_2$, respectively, the details of which are given in subsection IV-B.2. From the two figures, we can first observe that the minimum secrecy rates of all methods increase as the maximum transmit power increases. Besides, we know the performance gap between the system with the IRS and the system without the IRS increases with the transmit power, which validates the advantages of the introduced IRS. Secondly, in the scenario of $\Phi = \Phi_1$, we observe that Algorithm 1 and Algorithm 2 have the similar performances and that both of them outperform the other baselines. This is because both of them can guarantee to converge to a local (global) optimum. Thirdly, we observe that the performance of the ZF based heuristic algorithm is worse than that of "Rand" baseline with K = 2 and better than that with K = 1. This is acceptable due to the complexity of the "Rand" baseline is still higher than the ZF based heuristic algorithm. It also shows that the heuristic algorithm is more effective when K is small.



FIGURE 3. Minimum secrecy rate versus the number of reflecting elements on the IRS: M = 5, K = N = 2, P = 10 dB and $\kappa_h = 1$ with $h \in \mathcal{H}$.

Figure 3 plots the minimum secrecy rate versus the number of reflecting elements (antennas) L of the IRS. From this figure, we can observe that the minimum secrecy rates of all methods assisted by the IRS increase as the number of reflecting elements on the IRS increases, while the minimum secrecy rate of the system without the IRS remains constant. This is reasonable since a larger number of reflecting elements of the IRS can achieve higher array gain. This also validates the advantages of the introduced IRS for the studied systems. In addition, the performance gap between Algorithm 1 with $\Phi = \Phi_1$ and $\Phi = \Phi_3$ increases as Q decreases, especially for a large L. This is because the system with a large L requires a large Q to achieve more precise adjustment for the reflecting coefficients on the IRS. This indicates that in order to better achieve the array again brought by a larger L, we should use a larger Q for the proposed scheme with $\Phi = \Phi_3$.

Figure 4 shows the minimum secrecy rate versus the number of discrete reflecting coefficient values Q of the reflecting coefficients on the IRS. We can observe that the performances of the proposed algorithms with $\Phi = \Phi_1$, $\Phi = \Phi_2$ and $\Phi = \Phi_3$ decrease successively due to the fact $\Phi_3 \subseteq \Phi_2 \subseteq \Phi_1$. Furthermore, the performance of the proposed algorithm with $\Phi = \Phi_3$ increases as Q increases. This is because a larger Qallows a much finer adjustment to the reflecting coefficients



FIGURE 4. Minimum secrecy rate versus the number of reflecting coefficients of each reflecting elements of the IRS: M = L = 5, N = 2, P = 10 dB and $\kappa_h = 1$ with $h \in \mathcal{H}$.

on the IRS. Thus, the minimum secrecy rate can be improved. Finally, we observe that the proposed algorithm with $\Phi = \Phi_3$ and Q = 8 or 16 can achieve a similar performance to the proposed algorithm with $\Phi = \Phi_2$.



FIGURE 5. Minimum secrecy rate versus the number of legitimate users: M = 10, L = 5, N = 2, P = 10 dB and $\kappa_h = 3$ with $h \in \mathcal{H}$.

Figure 5 plots the minimum secrecy rate versus the number of legitimate receivers K. First, we observe that the minimum secrecy rate decrease as K increases, due to the fact that the beamforming gain and array gain need to be shared with more legitimate users. Moreover, its similar to Fig. 2, we can observe that the performance of "Algorithm 1" is better than that of "ZF based Heuristic Algorithm" and other baseline schemes. In addition, we also know that the low-complexity heuristic algorithm is more effective than the other suboptimal baseline schemes when K is small, especially for K = 1 or 2.

Figure 6 investigates the convergence performances of the proposed algorithms. For convenience, we first define Γ as the normalized performance gap between the values of the objective function in the two successive iterations of Algorithm 1 with $\Phi = \Phi_1$, with $\Phi = \Phi_2(1)$,



FIGURE 6. Convergence performance versus the number of iterations: M = 5, L = 5, K = 1, N = 1, P = 5 dB and $\kappa_h = 1$ with $h \in \mathcal{H}$.

and Algorithm 2. We can first observe that the objective functions in all methods increases with every iteration, which validates the convergence analysis studied in Section IV. Moreover, the convergence performance of Algorithm 1 with $\Phi = \Phi_2(1)$ is worse than that with $\Phi = \Phi_1$. This is because Algorithm 1 with $\Phi = \Phi_2(1)$ has the main drawback that there is no beforehand choice for the relaxation factor to speed up the convergence [32]. Besides, it is worth noting that the number of iterations to achieve convergence in Algorithm 2 is smaller than that in Algorithm 1. This is because in the each iteration of Algorithm 2, we optimize the original problem without approximation and provide the global optimal solution for each subproblem.

VII. CONCLUSIONS

In this paper, we have investigated the joint beamforming and reflecting coefficient designs for a programmable downlink MISO broadcast system with multiple eavesdroppers. In particular, considering the scenario that the channel responses of the legitimate receivers are highly correlated with those of the eavesdroppers, it is intractable to guarantee the secret communications with the use of beamforming only at the transceivers. Hence, we have explored the use of the IRS to create a programmable wireless environment by providing additional communication links to increase the SNR at the legitimate receivers while suppressing the SNR at the eavesdropper. Specifically, we have formulated a minimumsecrecy-rate maximization problem under various practical constraints on the reflecting coefficients, which captures the scenarios of both continuous and discrete reflecting coefficients of the reflecting elements. Since the formulated problem is a non-convex problem, we have proposed an efficient algorithm to solve it in an iterative manner and theoretically analyzed its convergence. In addition, we have developed two suboptimal algorithms with closed-form solutions to further reduce the complexity. Finally, the simulation results have validated the advantages of the IRS and the effectiveness of the proposed algorithms.

APPENDIX

A. PROOF OF THEOREM 1

We first prove the lower bound function of $f_k^{\rm B}(\boldsymbol{W}, \boldsymbol{v})$, and then prove the upper bound function of $f_{k,n}^{\rm E}(\boldsymbol{W}, \boldsymbol{v})$ in this part, which is similar to the proof in [32].

To obtain the lower bound function of $f_k^{\text{B}}(\boldsymbol{W}, \boldsymbol{v})$, we first prove the convexity of function $f(x, y) = -\ln(1 - |x|^2/y)$. Since $-\ln(1 - z)$ is an increasing and convex function with respect to z and $z = |x|^2/y$ is a convex function with respect to (x, y) in the domain $\{(x, y) | 0 \le y \le |x|^2\}$, f(x, y) is thus a convex function. According to the first-order Taylor series expansion of f(x, y) at (\tilde{x}, \tilde{y}) , we have

$$f(x,y) \ge f(\tilde{x},\tilde{y}) + \nabla_{\tilde{x}}f(x,\tilde{y})(x-\tilde{x}) + \nabla_{\tilde{y}}f(\tilde{x},y)(y-\tilde{y})$$

$$= f(\tilde{x},\tilde{y}) + 2\frac{\Re\{\tilde{x}(x-\tilde{x})\}}{\tilde{y} - |\tilde{x}|^2} - \frac{|\tilde{x}|^2}{\tilde{y}\left(\tilde{y} - |\tilde{x}|^2\right)}(y-\tilde{y}).$$

(54)

Then, setting $b = y - |x|^2$ and $\tilde{b} = \tilde{y} - |\tilde{x}|^2$, we have

$$\ln(1 + \frac{|x|^2}{b}) \ge \ln(1 + \frac{|\tilde{x}|^2}{\tilde{b}}) + 2\frac{\Re\{\tilde{x}x\}}{\tilde{b}} - \frac{|\tilde{x}|^2}{\tilde{b}(\tilde{b} + |\tilde{x}|^2)}(b + |x|^2) - \frac{|\tilde{x}|^2}{\tilde{b}}.$$
 (55)

Finally, letting $x = v^H H_k w_k$, $b = b_k (W, v)$, $\tilde{x} = (v^{(t)})^H H_k w_k^{(t)}$ and $\tilde{b} = b_k (W^{(t)}, v^{(t)})$, we can obtain (20). To obtain the lower bound function of $f_k^{\rm B} (W, v)$, since

To obtain the lower bound function of $f_k^{\rm B}(\boldsymbol{W}, \boldsymbol{\nu})$, since function $\ln(1 + z)$ is concave function with respect to z, we have

$$\ln(1+z) \le \ln(1+\bar{z}) + (z-\bar{z})/(1+\bar{z}).$$
(56)

Then, we have

$$f_{k,n}^{\mathrm{E}}(\boldsymbol{W},\boldsymbol{\theta}) \leq f_{k,n}^{\mathrm{E}}(\boldsymbol{W}^{(t)},\boldsymbol{v}^{(t)}) + (1 + \frac{\left| (\boldsymbol{v}^{(t)})^{H} \boldsymbol{G}_{n} \boldsymbol{w}_{k}^{(t)} \right|^{2}}{q_{k,n}(\boldsymbol{W}^{(t)},\boldsymbol{v}^{(t)})})^{-1} \times (\frac{\left| \boldsymbol{v}^{H} \boldsymbol{G}_{n} \boldsymbol{w}_{k} \right|^{2}}{q_{k,n}(\boldsymbol{W},\boldsymbol{v})} - \frac{\left| (\boldsymbol{v}^{(t)})^{H} \boldsymbol{G}_{n} \boldsymbol{w}_{k}^{(t)} \right|^{2}}{q_{k,n}(\boldsymbol{W}^{(t)},\boldsymbol{v}^{(t)})}).$$
(57)

Then, since it is straightforward to know $q_{k,n}(\boldsymbol{W}, \boldsymbol{v}) \geq q_{k,n}(\boldsymbol{W}, \boldsymbol{v}; \boldsymbol{W}^{(t)}, \boldsymbol{v}^{(t)})$, we have $\frac{|\boldsymbol{v}^H \boldsymbol{G}_n \boldsymbol{w}_k|^2}{q_{k,n}(\boldsymbol{W}, \boldsymbol{v})} \leq \frac{|\boldsymbol{v}^H \boldsymbol{G}_n \boldsymbol{w}_k|^2}{q_{k,n}(\boldsymbol{W}, \boldsymbol{v}; \boldsymbol{W}^t, \boldsymbol{v}^t)}$. Finally, we have (21).

B. PROOF OF THEOREM 2

As aforementioned, the sequence of $(\mathbf{W}^{(t)}, \mathbf{v}^{(t)})$ will converges to $(\mathbf{W}^*, \mathbf{v}^*)$ as $t \to \infty$. Next, we write the Lagrangian function of $(\mathbf{P}2-t)$ as

$$\mathcal{L}(\boldsymbol{W}, \boldsymbol{v}, \boldsymbol{\eta}, \boldsymbol{\chi}) = \mathcal{R}^{\text{lb}}(\boldsymbol{W}^{(t)}, \boldsymbol{v}^{(t)}; \boldsymbol{W}^{(t)}, \boldsymbol{v}^{(t)}) \\ + \sum_{l=1}^{L+1} \eta_l (1 - v_l) + \boldsymbol{\chi} (P - \sum_{k=1}^K \|\boldsymbol{w}_k\|^2),$$
(58)

where $\eta = [\eta_1, ., \eta_{L+1}] \in \mathbb{C}^{1 \times (L+1)}$ and χ are the dual variables. Then, when $t \to \infty$, the corresponding KKT conditions are given as follows:

$$\nabla_{\boldsymbol{\nu}_n} \mathcal{R}^{\text{lb}}(\boldsymbol{W}^*, \boldsymbol{\nu}^*; \boldsymbol{W}^*, \boldsymbol{\nu}^*) - \eta_l^* = 0, \quad \forall l, \qquad (59)$$

$$\eta_l^*(1 - v_l^*) = 0, \quad \forall l,$$
 (61)

$$\chi^*(P - \sum_{k=1}^K \|\boldsymbol{w}_k^*\|^2) = 0.$$
 (62)

According to (23), and notice that W^* and v^* are the optimal solutions of the convex optimization problems of (P3-A) and (P3 - B), respectively, we know the above KKT conditions are all satisfied. Hence, the converged solution (W^* , v^*) is a KKT point.

C. PROOF OF THEOREM 3

Firstly, (40) can be rewritten as follows:

$$\max_{V \ge 0} \ln(1 + \bar{\boldsymbol{h}}_1^H \boldsymbol{V} \bar{\boldsymbol{h}}_1) - \operatorname{Tr}((\operatorname{diag}(\boldsymbol{\chi}) + y(\bar{\boldsymbol{g}}_1 \bar{\boldsymbol{g}}_1^H))\boldsymbol{V}). \quad (63)$$

Let us define

$$\tilde{\boldsymbol{V}} = (\operatorname{diag}(\boldsymbol{\chi}) + y(\bar{\boldsymbol{g}}_1 \bar{\boldsymbol{g}}_1^H))^{0.5} \boldsymbol{V}(\operatorname{diag}(\boldsymbol{\chi}) + y(\bar{\boldsymbol{g}}_1 \bar{\boldsymbol{g}}_1^H))^{0.5}, \quad (64)$$

then, (63) can be rewritten as

$$\max_{\tilde{V} \succeq 0} \ln(1 + \varepsilon^2 \boldsymbol{c}^H \tilde{V} \boldsymbol{c}) - \operatorname{Tr}(\tilde{V}), \tag{65}$$

where

$$\boldsymbol{c} = 1/\varepsilon (\operatorname{diag}(\boldsymbol{\chi}) + y(\bar{\boldsymbol{g}}_1 \bar{\boldsymbol{g}}_1^H))^{-0.5} \bar{\boldsymbol{h}}_1, \qquad (66)$$

$$\varepsilon = \left\| (\operatorname{diag}(\boldsymbol{\chi}) + y(\bar{\boldsymbol{g}}_1 \bar{\boldsymbol{g}}_1^H))^{-0.5} \bar{\boldsymbol{h}}_1 \right\|.$$
(67)

It is straightforward to know that the rank of the optimal V must be one [46], which can be represented as $\tilde{V} = w \tilde{v} \tilde{v}^H$ where $\|\tilde{v}\|^2 = 1$. Then, from (65), the optimal \tilde{v}^* is *c*, and *w* can be obtained by solve the following problem:

$$\max_{w\geq 0} \ln(1+\varepsilon^2 w) - w. \tag{68}$$

Obviously, the optimal w^* is $\left(1 - \frac{1}{\varepsilon^2}\right)^+$. Thus, we have proved this theorem.

REFERENCES

- F. Rusek, D. Persson, B. K. Lau, E. G. Larsson, T. L. Marzetta, O. Edfors, and F. Tufvesson, "Scaling up MIMO: Opportunities and challenges with very large arrays," *IEEE Signal Process. Mag.*, vol. 30, no. 1, pp. 40–60, Jan. 2013.
- [2] A. Nosratinia, T. E. Hunter, and A. Hedayat, "Cooperative communication in wireless networks," *IEEE Commun. Mag.*, vol. 42, no. 10, pp. 74–80, Oct. 2004.
- [3] Y.-C. Liang, K.-C. Chen, G. Y. Li, and P. Mahonen, "Cognitive radio networking and communications: An overview," *IEEE Trans. Veh. Technol.*, vol. 60, no. 7, pp. 3386–3407, Sep. 2011.
- [4] C. Liaskos, S. Nie, A. Tsioliaridou, A. Pitsillides, S. Ioannidis, and I. Akyildiz, "A new wireless communication paradigm through software-controlled metasurfaces," *IEEE Commun. Mag.*, vol. 56, no. 9, pp. 162–169, Sep. 2018.
- [5] H. Yang, X. Cao, F. Yang, J. Gao, S. Xu, M. Li, X. Chen, Y. Zhao, Y. Zheng, and S. Li, "A programmable metasurface with dynamic polarization, scattering and focusing control," *Sci. Rep.*, vol. 6, Oct. 2016, Art. no. 035692.
- [6] T. J. Cui, M. Q. Qi, X. Wan, J. Zhao, and Q. Cheng, "Coding metamaterials, digital metamaterials and programmable metamaterials," *Light, Sci. Appl.*, vol. 3, no. 10, p. e218, Oct. 2014.
- [7] G. W. Ford and W. H. Weber, "Electromagnetic interactions of molecules with metal surfaces," *Phys. Rep.*, vol. 113, no. 4, pp. 195–287, Nov. 1984.
- [8] R. Zhang, Y.-C. Liang, C. C. Chai, and S. Cui, "Optimal beamforming for two-way multi-antenna relay channel with analogue network coding," *IEEE J. Sel. Areas Commun.*, vol. 27, no. 5, pp. 699–712, Jun. 2009.

- [9] S. Hu, F. Rusek, and O. Edfors, "Beyond massive MIMO: The potential of data transmission with large intelligent surfaces," *IEEE Trans. Signal Process.*, vol. 66, no. 10, pp. 2746–2758, May 2018.
- [10] G. Yang, Y.-C. Liang, R. Zhang, and Y. Pei, "Modulation in the air: Backscatter communication over ambient OFDM carrier," *IEEE Trans. Commun.*, vol. 66, no. 3, pp. 1219–1233, Mar. 2018.
- [11] D. Li and Y.-C. Liang, "Adaptive ambient backscatter communication systems with MRC," *IEEE Trans. Veh. Technol.*, vol. 67, no. 12, pp. 12352–12357, Dec. 2018.
- [12] H. Guo, Q. Zhang, S. Xiao, and Y. C. Liang, "Exploiting multiple antennas for cognitive ambient backscatter communication," *IEEE Internet Things J.*, vol. 6, no. 1, pp. 765–775, Feb. 2019.
- [13] R. Long, H. Guo, L. Zhang, and Y.-C. Liang, "Full-duplex backscatter communications in symbiotic radio systems," *IEEE Access*, vol. 7, pp. 21597–21608, 2019.
- [14] G. Yang, Q. Zhang, and Y.-C. Liang, "Cooperative ambient backscatter communications for green Internet-of-Things," *IEEE Internet Things J.*, vol. 5, no. 2, pp. 1116–1130, Apr. 2018.
- [15] Q. Zhang, H. Guo, Y.-C. Liang, and X. Yuan, "Constellation learningbased signal detection for ambient backscatter communication systems," *IEEE J. Sel. Areas Commun.*, vol. 37, no. 2, pp. 452–463, Feb. 2019.
- [16] Q. Wu and R. Zhang, "Intelligent reflecting surface enhanced wireless network: Joint active and passive beamforming design," in *Proc. IEEE Global Commun. Conf. (GLOBECOM)*, Dec. 2018, pp. 1–6.
- [17] Y. Han, W. Tang, S. Jin, C.-K. Wen, and X. Ma, "Large intelligent surface-assisted wireless communication exploiting statistical CSI," 2018, arXiv:1812.05429. [Online]. Available: https://arxiv.org/abs/1812.05429
- [18] X. Tan, Z. Sun, D. Koutsonikolas, and J. M. Jornet, "Enabling indoor mobile millimeter-wave networks based on smart reflect-arrays," in *Proc. IEEE INFOCOM*, Apr. 2018, pp. 270–278.
- [19] X. Tan, Z. Sun, J. M. Jornet, and D. Pados, "Increasing indoor spectrum sharing capacity using smart reflect-array," in *Proc. IEEE ICC*, May 2016, pp. 1–6.
- [20] Q. Wu and R. Zhang, "Beamforming optimization for intelligent reflecting surface with discrete phase shifts," in *Proc. IEEE ICASSP*, May 2019, pp. 7830–7833.
- [21] C. Huang, A. Zappone, G. C. Alexandropoulos, M. Debbah, and C. Yuen, "Reconfigurable intelligent surfaces for energy efficiency in wireless communication," 2018, arXiv:1810.06934. [Online]. Available: https://arxiv.org/abs/1810.06934
- [22] C. Huang, G. C. Alexandropoulos, A. Zappone, M. Debbah, and C. Yuen, "Energy efficient multi-user MISO communication using low resolution large intelligent surfaces," in *Proc. IEEE Globecom Workshops (GC Wkshps)*, Dec. 2018, pp. 1–6.
- [23] Q.-U.-A. Nadeem, A. Kammoun, A. Chaaban, M. Debbah, and M.-S. Alouini, "Asymptotic analysis of large intelligent surface assisted MIMO communication," 2019, arXiv:1903.08127. [Online]. Available: https://arxiv.org/abs/1903.08127
- [24] Y. Pei, Y.-C. Liang, K. C. Teh, and K. H. Li, "Secure communication in multiantenna cognitive radio networks with imperfect channel state information," *IEEE Trans. Signal Process.*, vol. 59, no. 4, pp. 1683–1693, Apr. 2011.
- [25] A. D. Wyner, "The wire-tap channel," *Bell Syst. Tech. J.*, vol. 54, no. 8, pp. 1355–1387, 1975.
- [26] I. Csiszar and J. Korner, "Broadcast channels with confidential messages," *IEEE Trans. Inf. Theory*, vol. IT-24, no. 3, pp. 339–348, May 1978.
- [27] S. Leung-Yan-Cheong and M. E. Hellman, "The Gaussian wire-tap channel," *IEEE Trans. Inf. Theory*, vol. IT-24, no. 4, pp. 451–456, Jul. 1978.
- [28] Y. Liang, G. Kramer, H. V. Poor, and S. Shamai, "Compound wiretap channels," *EURASIP J. Wireless Commun. Netw.*, vol. 2009, no. 1, pp. 1– 12, Mar. 2009.
- [29] Y. Pei, Y.-C. Liang, L. Zhang, K. C. Teh, and K. H. Li, "Secure communication over MISO cognitive radio channels," *IEEE Trans. Wireless Commun.*, vol. 9, no. 4, pp. 1494–1502, Apr. 2010.
- [30] K. Cumanan, Z. Ding, B. Sharif, G. Y. Tian, and K. K. Leung, "Secrecy rate optimizations for a MIMO secrecy channel with a multiple-antenna eavesdropper," *IEEE Trans. Veh. Technol.*, vol. 63, no. 4, pp. 1678–1690, May 2014.
- [31] L. Liu, R. Zhang, and K.-C. Chua, "Secrecy wireless information and power transfer with MISO beamforming," *IEEE Trans. Signal Process.*, vol. 62, no. 7, pp. 1850–1863, Apr. 2014.
- [32] A. A. Nasir, H. D. Tuan, T. Q. Duong, and H. V. Poor, "Secrecy rate beamforming for multicell networks with information and energy harvesting," *IEEE Trans. Signal Process.*, vol. 65, no. 3, pp. 677–689, Feb. 2017.

- [33] M. Zaslavskiy, F. Bach, and J. P. Vert, "A path following algorithm for the graph matching problem," *IEEE Trans. Pattern Anal. Mach. Intell.*, vol. 31, no. 12, pp. 2227–2242, Dec. 2009.
- [34] Z. Sheng, H. D. Tuan, A. A. Nasir, T. Q. Duong, and H. V. Poor, "Power allocation for energy efficiency and secrecy of wireless interference networks," *IEEE Trans. Wireless Commun.*, vol. 17, no. 6, pp. 3737–3751, Jun. 2018.
- [35] M. Grant, S. Boyd, and Y. Ye. CVX: MATLAB Software for Disciplined Convex Programming. Accessed: Jun. 2009. [Online]. Available: http://stanford.edu/~boyd/cvx
- [36] J. Chen, Z. Zhang, Y.-C. Liang, X. Kang, and R. Zhang, "Resource allocation for wireless-powered IoT networks with short packet communication," *IEEE Trans. Wireless Commun.*, vol. 18, no. 2, pp. 1447–1461, Feb. 2019.
- [37] Y. Nesterov and A. Nemirovskii, Interior-Point Polynomial Algorithms in Convex Programming, vol. 13. Philadelphia, PA, USA: SIAM, 1994.
- [38] L. Zhang, R. Zhang, Y.-C. Liang, Y. Xin, and H. V. Poor, "On Gaussian MIMO BC-MAC duality with multiple transmit covariance constraints," *IEEE Trans. Inf. Theory*, vol. 58, no. 4, pp. 2064–2078, Apr. 2012.
- [39] S. Shafiee and S. Ulukus, "Achievable rates in Gaussian MISO channels with secrecy constraints," in *Proc. IEEE ISIT*, Jun. 2007, pp. 2466–2470.
- [40] J. Chen, Y.-C. Liang, and G. Yang, "A unified beamforming structure for wireless multicasting and power transfer in MIMO systems," in *Proc. IEEE ICCS*, Dec. 2016, pp. 1–6.
- [41] J. Jose, N. Prasad, M. Khojastepour, and S. Rangarajan, "On robust weighted-sum rate maximization in MIMO interference networks," in *Proc. IEEE ICC*, Jun. 2011, pp. 1–6.
- [42] S. Boyd and L. Vandenberghe, *Convex Optimization*. New York, NY, USA: Cambridge Univ. Press, 2004.
- [43] W. Yu and R. Lui, "Dual methods for nonconvex spectrum optimization of multicarrier systems," *IEEE Trans. Commun.*, vol. 54, no. 7, pp. 1310–1322, Jul. 2006.
- [44] D. W. H. Cai, T. Q. S. Quek, and C. W. Tan, "A unified analysis of maxmin weighted SINR for MIMO downlink system," *IEEE Trans. Signal Process.*, vol. 59, no. 8, pp. 3850–3862, Aug. 2011.
- [45] M. Schubert and H. Boche, "Solution of the multiuser downlink beamforming problem with individual SINR constraints," *IEEE Trans. Veh. Technol.*, vol. 53, no. 1, pp. 18–28, Jan. 2004.
- [46] R. Zhang, "Cooperative multi-cell block diagonalization with per-basestation power constraints," *IEEE J. Sel. Areas Commun.*, vol. 28, no. 9, pp. 1435–1445, Dec. 2010.



YING-CHANG LIANG (F'11) was a Professor with the University of Sydney, Australia, a Principal Scientist and Technical Advisor with the Institute for Infocomm Research (I2R), Singapore, and a Visiting Scholar with Stanford University, USA. He is currently a Professor with the University of Electronic Science and Technology of China (UESTC), China, where he leads the Center for Intelligent Networking and Communications (CINC) and serves as the Deputy Director of the

Artificial Intelligence Research Institute. His research interest include wireless networking and communications, cognitive radio, dynamic spectrum access, the Internet-of-Things, artificial intelligence, and machine learning techniques.

Dr. Liang was a Fellow of the IEEE for contributions to cognitive radio communications, and was also recognized by Thomson Reuters (now Clarivate Analytics) as a Highly Cited Researcher, from 2014 to 2018. He received the Prestigious Engineering Achievement Award from the Institute of Engineers Singapore, in 2007, the Outstanding Contribution Appreciation Award from the IEEE Standards Association, in 2011, and the Recognition Award from the IEEE Communications Society Technical Committee on Cognitive Networks, in 2018. He has also received numerous paper awards, with the recent ones, including the IEEE ICC Best Paper Award, in 2017, the IEEE ComSocs TAOS Best Paper Award, in 2016, and the IEEE Jack Neubauer Memorial Award, in 2014. He was the Chair of the IEEE Communications Society Technical Committee on Cognitive Networks. He served as the TPC Chair and the Executive Co-Chair of IEEE Globecom'17. He is the Founding Editor-in-Chief of the IEEE JOURNAL ON SELECTED AREAS IN COMMUNICATIONS-Cognitive Radio Series, and a Key Founder and now Editor-in-Chief of the IEEE TRANSACTIONS ON COGNITIVE COMMUNICATIONS AND NETWORKING. He served as a Guest/Associate Editor of the IEEE TRANSACTIONS ON WIRELESS COMMUNICATIONS, the IEEE JOURNAL OF SELECTED AREAS IN COMMUNICATIONS, the IEEE Signal Processing Magazine, the IEEE TRANSACTIONS ON VEHICULAR TECHNOLOGY, and the IEEE TRANSACTIONS ON SIGNAL AND INFORMATION PROCESSING OVER NETWORK. He was also an Associate Editor-in-Chief of the World Scientific Journal on Random Matrices: Theory and Applications. He was a Distinguished Lecturer of the IEEE Communications Society and the IEEE Vehicular Technology Society.



YIYANG PEI (S'09–M'12) received the B.Eng. and Ph.D. degrees in electrical and electronic engineering from Nanyang Technological University, Singapore, in 2007 and 2012, respectively. She was a Research Scientist with the Institute for Infocomm Research, Singapore, from 2012 to 2016. She is currently an Assistant Professor with the Singapore Institute of Technology. Her current research interest includes the application of machine learning to wireless communications and cognitive radio networks.



JIE CHEN received the B.S. degree in communication engineering from the Chongqing University of Posts and Telecommunications, China, in 2016. He is currently pursuing the Ph.D. degree with the University of Electronic Science and Technology of China (UESTC), China. His research interests include convex optimization, resource allocation in wireless communication networks, intelligent reflecting surface, and short packet communications.



HUAYAN GUO (M'18) received the B.S. degree in electronic information science and technology from the Beijing University of Posts and Telecommunications, Beijing, China, in 2012, and the Ph.D. degree from Peking University, Beijing, in 2017. He is currently a Postdoctoral Researcher with the Center for Intelligent Communications and Networking (CINC), University of Electronic Science and Technology of China. His research interests include cognitive radio and the Internet-of-Things communications.

• • •